

GOPEN ACCESS

Citation: Yazdi HP, Olito C, Kawakami T, Unneberg P, Schou MF, Cloete SWP, et al. (2023) The evolutionary maintenance of ancient recombining sex chromosomes in the ostrich. PLoS Genet 19(6): e1010801. https://doi.org/10.1371/journal.pgen.1010801

Editor: Giovanni Bosco, Geisel School of Medicine at Dartmouth, UNITED STATES

Received: December 12, 2022

Accepted: May 28, 2023

Published: June 30, 2023

Copyright: © 2023 Yazdi et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All the scripts and data required to reproduce the statistical results and figures in the manuscript are located in the public Github repository at https://github.com/ Homap/ostrich_PAR_analysis. The raw sequencing files are available in the NCBI Short Read Archive (SRA) under BioProject ID PRJNA952688 with SRA accession numbers (SRR24085072, SRR24085071, SRR24085070, SRR24085069, SRR24085065, SRR24085067, SRR24085066, SRR24085065, SRR24085064, SRR24085063). The variant calls used for this study are deposited **RESEARCH ARTICLE**

The evolutionary maintenance of ancient recombining sex chromosomes in the ostrich

Homa Papoli Yazdi¹[•]*, Colin Olito¹[•], Takeshi Kawakami^{2,3}, Per Unneberg⁴, Mads F. Schou¹, Schalk W. P. Cloete^{5,6}, Bengt Hansson¹, Charlie K. Cornwallis¹

 Department of Biology, Lund University, Lund, Sweden, 2 Science for Life Laboratory, Department of Cell and Molecular Biology, Uppsala University, Uppsala, Sweden, 3 Embark Veterinary, Inc., Boston, Massachusetts, United States of America, 4 Department of Cell and Molecular Biology, National Bioinformatics Infrastructure Sweden, Science for Life Laboratory, Uppsala University, Uppsala, Sweden, 5 Directorate Animal Sciences, Western Cape Department of Agriculture, Elsenburg, South Africa,

6 Department of Animal Sciences, Stellenbosch University, Matieland, South Africa

So These authors contributed equally to this work.

* Homa.Papoli_Yazdi@biol.lu.se

Abstract

Sex chromosomes have evolved repeatedly across the tree of life and often exhibit extreme size dimorphism due to genetic degeneration of the sex-limited chromosome (e.g. the W chromosome of some birds and Y chromosome of mammals). However, in some lineages, ancient sex-limited chromosomes have escaped degeneration. Here, we study the evolutionary maintenance of sex chromosomes in the ostrich (Struthio camelus), where the W remains 65% the size of the Z chromosome, despite being more than 100 million years old. Using genome-wide resequencing data, we show that the population scaled recombination rate of the pseudoautosomal region (PAR) is higher than similar sized autosomes and is correlated with pedigree-based recombination rate in the heterogametic females, but not homogametic males. Genetic variation within the sex-linked region (SLR) (π = 0.001) was significantly lower than in the PAR, consistent with recombination cessation. Conversely, genetic variation across the PAR (π = 0.0016) was similar to that of autosomes and dependent on local recombination rates, GC content and to a lesser extent, gene density. In particular, the region close to the SLR was as genetically diverse as autosomes, likely due to high recombination rates around the PAR boundary restricting genetic linkage with the SLR to only ~50Kb. The potential for alleles with antagonistic fitness effects in males and females to drive chromosome degeneration is therefore limited. While some regions of the PAR had divergent male-female allele frequencies, suggestive of sexually antagonistic alleles, coalescent simulations showed this was broadly consistent with neutral genetic processes. Our results indicate that the degeneration of the large and ancient sex chromosomes of the ostrich may have been slowed by high recombination in the female PAR, reducing the scope for the accumulation of sexually antagonistic variation to generate selection for recombination cessation.

in the Dryad Database (https://doi.org/10.5061/ dryad.pnvx0k6sx).

Funding: The authors would like to acknowledge support from Science for Life Laboratory (Wallenberg Advanced Bioinformatic Infrastructure project 1502). The computations were enabled by resources in project SNIC2020-16-269 provided by the Swedish National Infrastructure for Computing (SNIC) at UPPMAX, partially funded by the Swedish Research Council through grant agreement no. 2018-05973. The resource population sampled was managed under the auspices of the Western Cape Department of Agriculture and partially funded by the Western Cape Agricultural Research Trust (Reference No: 0070/000 Volstruise). This research was funded by the Swedish Research Council (grant number 2017-03880 to CKC; 2016-00689 to BH), The Wenner-Gren and Olle Engkvist Foundations (postdoctoral fellowships to CO) and Knut and Alice Wallenberg Foundation (Wallenberg Academy fellowship numbers 2013.0129 & 2018.0138 to CKC) and Carl Tryggers Foundation (grant numbers 12:92 & 19:71 to CKC). PU is financially supported by the Knut and Alice Wallenberg Foundation as part of the National Bioinformatics Infrastructure Sweden at Science for Life Laboratory. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Author summary

In birds, sex is determined by a pair of sex chromosomes where females are ZW and males are ZZ. In many taxa, the W chromosome has degenerated except for a small part that continues to recombine in both sexes called the pseudoautosomal region (PAR). The prominent theory for why recombination suppression between Z and W evolves is that selection favours linkage between the sex-determination locus and alleles with sex-specific fitness effects. Recombination cessation results in reduced selection efficiency and ultimately the degeneration of the W chromosome. There are, however, some ancient sex chromosomes that deviate from this path of W demise that present one of the fascinating puzzles of sex chromosome evolution. In ostriches, sex chromosomes with a long PAR have been maintained for over 100 million years. Here, we show that high recombination between the sex-linked region and PAR loci. This suggests that the integrity of sex chromosomes may be maintained in species where high recombination rates minimize the influence of sex linkage on neighboring genes, reducing the likelihood that mutations with sex-specific fitness effects accumulate.

Introduction

In many taxa, sex is determined by genes residing on a pair of homologous chromosomes, such as the XY chromosomes of mammals and the ZW chromosomes of birds [1]. These sex chromosomes typically have two regions, a sex-linked region (SLR) where sex-determining genes are located and recombination is suppressed, and one or more pseudoautosomal regions (PAR) where recombination persists to ensure proper chromosome pairing during meiosis (Fig.1) [2,3]. Interestingly, the size of the PAR differs widely among even closely related species, demonstrating the dynamic evolution of recombination on sex chromosomes [3]. However, it remains unclear why some species have sex chromosomes with small PARs and extensive non-recombining regions [4–6], while others have large PARs that are maintained over long evolutionary periods [7–9].

A popular explanation for why recombination suppression spreads across sex chromosomes and PARs degrade, is based on sexually antagonistic selection [13]. Alleles that are beneficial when expressed in one sex, but deleterious when expressed in the other (i.e., sexually antagonistic variation) cause indirect selection for suppressed recombination between sex chromosomes. This results in female-beneficial alleles becoming associated with the W and male-beneficial alleles becoming associated with the Z [14]. Importantly, the conditions for maintaining such sexually antagonistic variation are more permissive when loci are partially genetically linked to the SLR [13]. At the same time, stronger sex-linkage increases the coalescence times of sampled alleles from the two sex chromosomes. This leads to increased neutral diversity at partially sex-linked sites within the PAR [15], and increases the potential for the accumulation of sexually antagonistic genetic variation [16].

A key process determining levels of genetic linkage between PAR loci and the SLR is the rate of recombination. High recombination rates close to the SLR are expected to reduce the proportion of PAR loci with sex-biased inheritance and the potential for sexually antagonistic variation to accumulate. Recombination rates can also impact other processes that influence the evolution of genetic variation on sex chromosomes [17]. For example, regions with low recombination, particularly when enriched in functional elements under selection, can show large reductions in neutral genetic diversity due to the impact of selection at linked sites



Fig 1. The structure of the ostrich Z chromosome. The Z chromosome consists of the PAR and the SLR. In the PAR, sex chromosomes recombine in both males (Z/Z) and females (Z/W), while in the SLR recombination only occurs in males. The cumulative recombination frequency between the PAR boundary (at ~52.2 Mb) and loci within the PAR determines the extent of sex-linked inheritance, and is calculated using the female and male genetic map lengths (80.628 cM and 42.641 cM, respectively from [10]) using the Kosambi map function [11]. Gray bars represent genes along the Z chromosome. Red bars indicate homologous genes that are still present within the SLR on the W chromosome (i.e., gametologous genes). *DMRT1* is the Z-linked avian sex-determining gene [12] (blue bar).

https://doi.org/10.1371/journal.pgen.1010801.g001

[18,19]. In contrast, regions with high recombination rates will be impacted by GC-biased gene conversion which can also resemble the effect of selection [20–22]. Quantifying recombination rates across the PAR in relation to the SLR is therefore crucial to understand the potential for sexually antagonistic selection, patterns of neutral genetic diversity and the evolution of expanded non-recombining regions.

Molecular and population genetic studies of species with highly degenerated W/Y chromosomes have revealed several characteristics of small PARs. In particular, the heterogametic sex (ZW females or XY males) typically has extreme recombination rates. In collared flycatchers (*Ficedula albicollis*), the PAR is only 630 Kb (1.05% of the Z length) and has a female recombination rate that is about 200 times higher than the genome-wide average, resulting in a GCrich sequence due to GC-biased gene conversion [5]. Similarly, in humans, PAR1 is about 2.7 Mb (1.73% of the X length) with a male recombination rate that is ~17 times higher than the autosomal average [23]. The role of sexual antagonism in the expansion of the SLR in these systems has been investigated by estimating the divergence in allele frequencies between males and females and examining sex-biased gene expression, which can be a sign of resolved sexual conflict [24]. In the human PAR1, there are divergent allele frequencies between the sex chromosomes and enrichment of male-biased expressed genes, but these patterns are not entirely explained by sexual antagonism [25,26]. However, in collared flycatchers, there is little evidence of male-female allele frequency divergence and sex-biased gene expression. This is potentially due to a recombination hotspot close to the PAR-SLR boundary reducing the opportunity for sexually antagonistic genetic variation to accumulate [5]. In contrast to these examples of small PARs, the population genetic dynamics of ancient sex chromosomes with large PARs remain largely unknown.

Sex chromosomes with long PARs are found in both animals [9,27,28] and dioecious plants [29]. Some of these sex chromosomes, such as those found in the flowering plant Silene latifolia [30], have evolved relatively recently and are in the early stages of degeneration. Sex chromosomes with long and ancient PARs are relatively rare, but notable examples occur within the avian lineage Palaeognathae (tinamous and ratites). In Palaeognathae, the sex chromosomes first evolved more than 100 MYA in a shared common ancestor with the Neognathae, which contains >99% of extant avian species [31]. Unlike the Neognathae, many Palaeognathae species, particularly the ratites, have retained long PARs ranging from between ~65% to 73% of the length of the Z chromosome [9,32-35]. For example, the PAR of the ostrich is ~52.2 Mb which comprises ~65% of the Z chromosome. Molecular evolution studies of the ostrich PAR have shown that rates of evolution, measured by synonymous substitution rates, are similar to autosomal sequences and that the PAR is not enriched for male-biased expressed genes [9]. Recombination frequency along the PAR obtained from an ostrich pedigree also showed a higher frequency of recombination along the PAR in females compared to males (Fig 1) [10]. However, the resolution of the markers along the Z chromosome was low with an average of 1 marker every ~250Kb. It is therefore unclear how the fine-scale recombination landscape of the ostrich PAR influences levels of genetic variation and the degree of genetic linkage with the SLR, a critical factor determining the scope for sexually antagonistic alleles to accumulate.

Here we use whole-genome re-sequencing and a genetic linkage map [10] of the ostrich sex chromosomes to test if recombination rate can help explain the evolutionary maintenance of the large, ancient PAR. First, we investigate the pattern of recombination across the sex chromosomes and its influence on the degree of genetic linkage with the SLR (measured by linkage disequilibrium (LD)). We also compare sex-averaged population recombination rates to male and female estimates from the genetic linkage map to quantify how historical recombination rates have been shaped by each sex. Second, we test how recombination rates influence genetic diversity and GC content and if this relates to gene density across the PAR. Third, we examine if there are regions of the PAR that potentially harbor sexually antagonistic alleles by calculating allele frequency differences between males and females. Finally, we evaluate whether patterns of genetic diversity and the observed differences in allele frequencies between males and females across the PAR, particularly in the region close to SLR, are consistent with predictions from population genetic theory using coalescent simulations based on [15].

Results

Recombination rate and linkage disequilibrium across the sex chromosomes

The sex-averaged population recombination rate ($\rho = 4N_er$) across the PAR (mean (SD): ρ / Kb = 0.17 (0.12)) was significantly higher than on similar sized autosomes (mean (SD): ρ / Kb = 0.13 (0.071)) (Mann-Whitney U: U = 40087, p = 5.867e-05). This is expected given cross-overs must occur in a smaller chromosomal segment of the female PAR. Recombination rates were highly variable across the sex chromosomes but could be divided into four regions with significantly different recombination rate regimes based on a change-point analysis (estimated change-points: 14.6, 48.1 and 53.6 Mb) (Figs 2A and S3 and S4 Table). The highest recombination rate at 2.2 Mb (ρ /Kb = 0.8) occurred at the distal end of the chromosomes, farthest from the PAR boundary. In the mid-PAR, there was a recombination valley containing the



Fig 2. Recombination rate and linkage disequilibrium (LD) across the ostrich Z chromosome. (A) Population scaled recombination rate ($\rho = 4N_er$) calculated from SNPs in windows of 200 Kb with 50 Kb overlap using LDhat [36] (Gray points). The horizontal dashed line in orange is the autosomal average for $\rho/Kb = 0.13$. The vertical dashed line in red at 2.2 Mb represents the maximum ρ/Kb across the PAR. The black dash-dotted line is the rolling average calculated over a span of ~1 Mb (B) Female, male and sex-averaged recombination rate (cM/Mb) in windows of 1 Mb obtained from a genetic linkage map [10]. The dash-dotted lines are the rolling averages calculated over a span of ~5 Mb (C) LD in windows of 200 Kb with 50 Kb overlap (Gray points). The horizontal dashed line in orange indicates the average autosomal LD ($r^2 = 0.15$). The black dash-dotted line is the rolling average calculated over a span of ~1 Mb. The triangular matrix plot indicates LD across the 200 Kb of the PAR boundary. The vertical dashed lines in A to C at ~52.2 Mb represents the PAR boundary. (D) Population scaled recombination rate (ρ/Kb) was significantly correlated to female (R = 0.35, p = 0.013) and sex-averaged (R = 0.32, p = 0.021) genetic map recombination rate (cM/Mb) across the PAR, but not to male recombination rate (R = 0.1, p = 0.45). Each dot represents an average for 1 Mb windows. (E) LD steeply declined with population scaled recombination rate (ρ/Kb) for the PAR and the SLR. (F) LD decay in relation to pairwise SNP distance across the SLR, mid-PAR (~23–28 Mb), autosomes, PAR-boundary (~47–52.2 Mb), the PAR and PAR-1-5 Mb.

https://doi.org/10.1371/journal.pgen.1010801.g002

minimum rate in the PAR at 22.3 Mb ($\rho/Kb = 0.02$). Examining the 5 Mb region immediately adjacent to the PAR boundary showed that the recombination rate was significantly higher than the autosomal average (mean (SD): $\rho/Kb = 0.19$ (0.05)) (Mann-Whitney U: U = 1463, p = 2.175e-06). In contrast, within the SLR the recombination rate dropped abruptly to an average of 0.1 ρ/Kb (SD = 0.089), consistent with a lack of recombination in females.

The recombination rate influenced LD patterns across the Z chromosome (Fig 2C). Regions with higher recombination rates had reduced levels of LD (Generalized Least Squares of LD~recombination rate: t = -8.1, p < 0.001), with LD declining rapidly with recombination rate (Fig 2E). The mean pairwise LD for the whole PAR was similar to that of autosomes (mean (SD) PAR = 0.132 (0.017), autosomes = 0.134 (0.017)) and reached a ~50 Kb pairwise SNP distance in the PAR and ~65 Kb in the autosomes. There was, however, a difference in LD decay between sections of the PAR according to their distance to the PAR boundary, consistent with changes in recombination rate (Fig 2F). Across the 200 Kb region spanning the PAR boundary, there was little LD between the PAR and SLR (Fig 2C), indicating that even PAR loci in close physical proximity to the PAR boundary (within ~50Kb) effectively segregate independently from the SLR.

The higher recombination rate in the regions closest to and farthest from the PAR boundary may be caused by cross-over pairing between the Z and W being forced into a smaller region in females than in males due to W chromosome degeneration [37]. To investigate whether females are driving the overall recombination rate in the PAR we used genetic map data from [10] (Fig 2B). Female recombination rate for the PAR (mean (SD) = 1.70 (1.8) cM/ Mb) was indeed significantly higher than the male rate (mean (SD) = 0.85 (0.93) cM/Mb, Mann-Whitney U: U = 1635, p = 0.025). Within the 5 Mb region closest to the PAR boundary, the female recombination reached 3.41 cM/Mb, while the male recombination rate was only 0.41 cM/Mb. Furthermore, ρ was significantly correlated with sex-averaged genetic map recombination rate (cM/Mb) across the PAR (R = 0.32, p = 0.02), and this was driven by female, not male recombination rate (Correlation with female map cM/Mb: R = 0.35, p = 0.013, Correlation with male map cM/Mb: R = 0.11, p = 0.45, Fig 2D). Together these results imply that historical recombination rates of the ostrich PAR are shaped by the recombination rate of heterogametic females.

Patterns of genetic diversity and female-male divergence in allele frequency across the sex chromosomes

Genetic diversity across the Z chromosome was variable with a clear break at the boundary between the PAR and the SLR (Fig 3A). The average level of genetic diversity in the PAR was similar to autosomal levels (mean (SD) π : PAR = 0.0016 (0.0004), autosomes = 0.0016 (0.0004), Mann-Whitney U: PAR vs autosomes, U = 97478, p = 0.1). In the SLR, the genetic diversity dropped abruptly (mean (SD) π = 0.001 (0.0002)), as expected with complete cessation of recombination (Fig 3A) and was significantly lower than within the PAR (Mann-Whitney U: PAR vs SLR, U = 5615, p = 1.349e-09). The heterogenous pattern of genetic diversity across the PAR was correlated with recombination rate (Fig 3F. GLS of genetic variation ~ ρ / Kb: t = 10.22, p < 0.001) and GC content (Fig 3B and 3G. GLS: t = 2.51, p = 0.013). This positive correlation with GC content is potentially due to GC-biased gene conversion during recombination events [20]. Conversely, genetic diversity was negatively related to gene density, although this relationship was weak (Fig 3C and 3H. GLS: t = -1.89, p = 0.06). These results are consistent with the idea that linked selection in regions with high gene density and low recombination rates leads to reduced genetic diversity [38,39].

We examined if there were regions of the PAR with allele frequency differences between females and males using two different measures, Tajima's D and F_{ST} between females and



Fig 3. Population genomic features were highly variable across the ostrich Z chromosome. (A) Pairwise nucleotide diversity (π), (B) GC content (%), (C) Coding DNA Sequence (CDS) density (%), (D) Tajima's D (T_d) and (E) Female-male F_{ST} (F_{FM}), calculated for 200-Kb non-overlapping windows (Gray points). Pairwise nucleotide diversity on the PAR plotted as a function of (F) population scaled recombination rate (p < 2.2e-16), (G) GC content (p = 0.0018) and (H) CDS density (p = 0.041). *p*-values are for GLS regression coefficients of each explanatory variable taken from the full model

reported in the text. In (A) to (E), the black dash-dotted line is the rolling average calculated over a span of ~1 Mb, the horizontal dashed line indicates the autosomal average and the vertical dashed line indicates the PAR-SLR boundary at ~52.2 Mb.

https://doi.org/10.1371/journal.pgen.1010801.g003

males ($F_{\rm FM}$). Partial linkage with the SLR causes neutral variants to have deeper coalescent times, resulting in a signature of elevated diversity resembling balancing selection, as indicated by higher values of Tajima's D [15]. We found Tajima's D was similar in the PAR and autosomes (Mann-Whitney U: U = 84963, p = 0.1), and not significantly higher at the PAR boundary, as expected if allele frequencies are not influenced by the SLR. Instead, Tajima's D was highest in the mid-PAR with the maximum value of 1.1 at 22.3 Mb (Fig 3D). Similarly, values of $F_{\rm FM}$ were highest mid-way through the PAR (Fig 3E). The average $F_{\rm FM}$ across the PAR was also significantly higher than both autosomes and the SLR (Mann-Whitney U: autosome– PAR, U = 65673, p < 0.001; PAR–SLR, U = 23558, p = 1.646e-08). The mid-PAR had the lowest recombination rate and genetic diversity, which might explain the elevated values of divergence in female-male allele frequency.

Are patterns across the PAR consistent with neutral genetic theory?

To test if the observed patterns of genetic diversity and female-male divergence across the PAR are consistent with neutral genetic processes, we used coalescent simulations based on [15]. Our results were broadly consistent with neutral genetic theoretical predictions of the expected average pairwise nucleotide diversity ($\bar{\pi}$), with empirical estimates falling within the 95% confidence intervals from coalescent simulations (Fig 4. See Materials & Methods: Coalescent simulations). Genetic diversity did, however, exceed the 95% confidence intervals in two 200 Kb windows, one at the start of the chromosome and one at the PAR boundary (Fig 4A). Focusing on the small region around the PAR boundary (~10–20 Kb), our simulations predicted a nonlinear increase in $\bar{\pi}$ (Fig 4C, solid line). This was not seen in our empirical estimates, where $\bar{\pi}$ was lower than expected adjacent to the PAR boundary (Fig 4C), and higher farther away.

Examining patterns of female-male divergence showed a region in the middle of the PAR (~15 to ~25 Mb) with elevated $F_{\rm FM}$ values that fell outside the 95% confidence interval of the simulations (Fig 4B). There was also a suggestive spike in $F_{\rm FM}$ very close to the PAR boundary (Fig 4D). Our simulations also predicted a sharp increase in $F_{\rm FM}$ in the small region adjacent to the PAR boundary, similar to patterns of genetic diversity. However, the windows with elevated $F_{\rm FM}$ values, that exceeded theoretical predictions, were too far away from the SLR to be sex-linked (between positions 20-30Mb on the Z). Overall, it appears that the high recombination rates in females, especially in the region adjacent to the PAR boundary (see Fig 2B), rapidly break down genetic associations between the SLR and PAR loci, resulting in empirical patterns of genetic diversity that are consistent with predictions from neutral theory.

Discussion

The maintenance of recombination in ancient sex chromosomes, as observed in ratites, is one of the long-standing conundrums of sex chromosome evolution. In this study, we used population genomic data from ostriches to investigate how patterns of recombination and genetic variation across the sex chromosomes change in relation to the SLR. We found that the genetic linkage between the SLR and the PAR was restricted to an extremely small region (~50Kb), which means that most of the PAR segregates independently from the SLR.

The PAR sequence that is tightly linked to the SLR is estimated to occur within a cumulative $\rho = 1$ from the PAR boundary in heterogametic females [15]. For the 200 Kb window adjacent



Fig 4. Predicting genetic diversity and female-male divergence across the ostrich PAR using neutral genetic theory. (A) Predicted average neutral genetic diversity, $\bar{\pi}$, and (B) female-male divergence, $F_{\rm FM}$, across the full PAR (Physical position 0–52.2 Mb). Solid lines indicate the mean value of 1,000 replicate coalescent simulations (Methods section "Coalescent simulations"), with shading indicating 95% confidence intervals (i.e., 95% of calculated values from the simulations fell within this interval). Points indicate empirical estimates calculated for 200 Kb windows. Panels (C) and (D) are high resolution illustrations (1 Kb windows) of the ~140 kb region immediately adjacent to the PAR boundary.

https://doi.org/10.1371/journal.pgen.1010801.g004

to the PAR the sex-averaged population scaled recombination rate was 127.1. This means that the region where $\rho = 1$ is predicted to be only ~1575 base pairs long. The length of this region is comparable to that found in *Silene latifolia* where only ~500 base pairs of the PAR were estimated to be in strong LD with the SLR [40]. Additionally, in our study, genetic variation was not above the autosomal mean, as expected if the PAR loci were in linkage with the SLR. While these results suggest that LD decays extremely fast as you move away from the PAR boundary, it is possible that the difference between the theoretical predictions from our simulations and our empirical estimates is due to low SNP densities near the PAR boundary, or possibly because there is an assembly gap of ~7.6 Kb between the PAR and the SLR (S1 Fig).

The recombination rates of the ostrich PAR and autosomes were comparable to studies on two other ratites. In the greater rhea (*Rhea americana*), the PAR is 52.5 Mb and the average recombination rate was similar to that of autosomal pairs 5 and 6 [35]. In the emu (*Dromaius*

novaehollandiae), a comparison of recombination rates between 14 PAR and 8 autosomal loci concluded that recombination rates are slightly higher, and LD slightly lower, in the PAR than the autosomes [41]. In this study, patterns of recombination were highly variable along the PAR. In addition to the effects of the PAR boundary, recombination variation was consistent with the possible localization of cross-overs towards the chromosome ends [42]. Interestingly, we found that variation in historical rates of recombination of the PAR was largely explained by patterns of recombination in females and not males (Fig 2D). A higher recombination rate in the heterogametic sex has been hypothesized to protect the PAR from degeneration, enabling the maintenance of long PARs over prolonged evolutionary time periods [3]. In the case of ostrich, it seems that the pairing of the W and Z also causes higher recombination rate in this region.

Another feature of the PAR was a recombination valley in the middle section which coincided with a reduction in genetic diversity, a positive Tajima's D and elevated levels of femalemale allelic divergence (Fig 3E). While the elevated female-male divergence might hint at the action of sexually antagonistic selection, coalescent simulations showed the observed patterns are still consistent with neutral predictions. We propose that this saddle-shaped pattern of recombination is consistent with the combined effects of: (i) enforced pairing between the Z and W chromosomes both at the distal end and close to the PAR boundary, and (ii) cross-over interference limiting recombination rates in the mid-PAR [37].

In systems with smaller PARs, such as humans [25] and collared flycatchers [5], there is little empirical support for the role of sexual antagonism in shaping recombination patterns of sex chromosomes (although confirming the role of sexually antagonistic selection is extremely challenging). Small PARs, however, have extreme recombination dynamics that make it difficult to confirm or deny the importance of sexually antagonistic polymorphisms in the degeneration of the sex-limited chromosome. Species with long PARs offer opportunities to study the influence of the SLR on sex chromosome evolution without such complications. In the large PAR of the recently evolved sex chromosomes of *Silene latifolia*, eight genes with positive Tajima's D were detected [40]. Two of these genes, that were closest to the SLR, had different female-male allele frequencies, but the other six genes that were loosely linked to the SLR did not show any sex differences in allele frequencies. A simulation study on these six genes concluded that the positive Tajima's D could not be explained solely by demography, and that sexually antagonistic selection might be responsible for creating the observed patterns under a scenario where they were historically closely linked to the SLR [43].

The recombination of ancient ratite sex chromosomes may be maintained by several processes, including sex-biased gene expression [3], a slower rate of molecular evolution [44], and a high recombination rate at the PAR boundary reducing LD between the SLR and PAR loci [3]. In ostriches, it is unlikely that the sex chromosomes are maintained due to sex-biased gene expression, as the genes on the PAR in adults are equally expressed in both sexes [45]. Ratites, including the ostrich, do however have a slower rate of molecular evolution [46] making it possible that this has decreased the accumulation of genetic mutations that degenerate the W chromosome. Our study also highlights that a high recombination rate at the PAR boundary might prevent the accumulation of sexually antagonistic mutations. The theory for the evolution of recombination suppression of sex chromosomes due to sexual antagonism relies on sexually antagonistic alleles building genetic associations with the SLR. A key insight from our results is that an elevated recombination rate at the PAR boundary can greatly restrict the size of the region where sexually antagonistic alleles can become genetically associated with the SLR, providing an explanation for the evolutionary maintenance of ancient recombining sex chromosomes.

Materials and methods

Ethics statement

All procedures were approved by the Departmental Ethics Committee for Research on Animals (DECRA) of the Western Cape Department of Agriculture, reference no. AP/BR/O/ SC14.

Study population, sampling, and sequencing

Blood samples of *Struthio camelus* were obtained from Western Cape Department of Agriculture's ostrich research facility in Oudtshoorn, South Africa. Since 1995, individuals have been bred in pairs at the research facility to create pedigrees. At the time of sampling, the pedigrees contained 1531 males and 2067 females. We selected 5 males and 5 females for sequencing using the program PedMine [47]. PedMine identifies individuals with most distant links within pedigrees allowing the maximum amount of genetic diversity in populations to be sampled. Samples were sequenced at Science for Life Laboratory, the National Genomics Infrastructure, using paired end with 126 base pairs on Illumina HiSeq 2500, following manufacturer's protocol.

Mapping, variant calling and filtering

We implemented a snakemake [48] workflow for mapping and variant calling. Briefly, reads were trimmed with cutadapt version 2.10 [49] and then mapped to the optical map improved reference genome (Struthio_camelus.20130116.OM.fa) with bwa version 0.7.17.r1188 [50]. Mean coverage per sample is presented in S1 Table. Duplicates were marked with Picard MarkDuplicate [51]. The ostrich Z chromosome in the assembly version used in this study consists of 12 scaffolds (S2 Table). By measuring the average male and female coverage, the coordinate of the PAR boundary was determined to be in superscaffold36 between 3516672 and 3524264 with gap size of 7592 nucleotides (S1 Fig). We checked for the existence of gametologous genes on the 8 SLR scaffolds by identifying annotated genes with a copy on a putatively W-linked scaffold. We measured the male to female coverage ratio for the putative W-linked scaffold and if the ratio was close to zero, we determined the scaffold to be W-specific and containing gametologous gene (S3 Table). Heterozygous SNPs in females overlapping with these genes were removed from further analyses since they reflect divergence between Z and W since recombination cessation.

Variant calling was performed with GATK version 4.1.4.1 following best practice procedures developed at the Broad Institute [52]. The GATK HaplotypeCaller was run individually on each sample to generate GVCF output. GVCF files for all samples were imported to a GenomicsDB datastore, followed by genotyping with GATK GenotypeGVCFs to produce a final raw variant call set. Several filtering steps were performed on the raw call set to obtain the final call set of high quality. Biallelic SNPs were selected with GAKT SelectVariants and filtered with GATK VariantFilteration using best practice options QUAL < 30, QualByDepth (QD) < 2.0, RMSMappingQuality (MQ) < 40.0, MappingQualityRankSumTest (MQRankSum) < -12.5, FisherStrand (FS) > 60.0, ReadPosRankSumTest < -8.0 and StrandOddsRatio (SOR) > 3.0. We removed variants overlapping with repeats annotated by the *aves* repeat library using BEDTools intersect [53]. We filtered SNPs with more than twice the average coverage (>70 reads) and less than 5 reads per site. SNPs in the SLR in females are expected to occur only as haploid. However, heterozygous SNPs in the SLR in females can occur either due to genotyping error or due to the divergence of the Z and W sequences in the gametologous region. We therefore filtered the heterozygous SNPs in females in the SLR. This left us with 5,776,166 SNPs for autosomes, 268,006 SNPs for the PAR and 89,540 SNPs for the SLR. Distribution of alternative allele frequency and per site depth is shown in S2 Fig. To filter the background non-variant sites, we calculated coverage per site using samtools version 1.14 [54]. We used the hard-masked reference genome for repeats and filtered sites that had a minimum of 5 reads or a maximum of 70 reads. Coverage filter removed 78,235,359 sites from the whole genome. VCF files are publicly available in Dryad database [55].

Measures of population scaled recombination and linkage disequilibrium

Pedigree-based recombination rate provides us with an estimate of the recombination rate for one generation, but population scaled recombination rate ($\rho = 4N_e r$) gives us an estimate of recombination in the history of sample. Population scaled recombination rate was estimated for each Z scaffold separately in windows of 1000 SNPs with an overlap of 200 SNPs. The interval program in LDhat 2.2 [36] was used, and three independent Markov-Chain Monte Carlo (MCMC) chains were run with a block penalty of 5 and 25 million iterations. We sampled the chain every 5000 iterations and discarded the first fifth (5,000,000 iterations) as burn-in. To determine where the trend in population scaled recombination rates changes, we performed change-point analysis using the segmented package in R [56]. The change-point analysis with sex-averaged population scaled recombination rate returned 3 significant change-points along the graph, at 14.6, 48.1 and 53.6 Mb (S3 Fig). We have used these change points to define four regions: the SLR where recombination rate drops to an average of 0.1; the 5 Mb segment closest to the PAR boundary where recombination is higher than autosomal average; the mid-PAR containing the minimum recombination rate at 22.3 Mb, and the region most distant from the PAR boundary where recombination frequency reaches its maximum value well above the autosomal average at 2.2 Mb (S4 Table). Pairwise linkage disequilibrium (LD) was measured as the square of the correlation coefficient between the allelic states (r^2) for all pairs of SNPs within 200 Kb window with 50 Kb overlap after filtering for Hardy-Weinberg equilibrium (HWE) in vcftools [57] using PopLDdecay [58].

Measures of genetic variation and female-male allelic differentiation

Pairwise nucleotide diversity (π), the number of segregating sites (θ) and the relationship between the two, measured as Tajima's D statistic [59] for neutrally evolving sequences were calculated across chromosome Z in 200 and 1000 Kb non-overlapping windows using vcftools version 1.16 [57] and custom Python scripts located under https://github.com/Homap/ ostrich_PAR_analysis/tree/main/code/analysis/diversity. All population genetics measures were calculated for the SLR considering its haploid state in females. We investigated the relationship between genetic variation in the PAR with recombination rate, GC content and gene density using Generalized Least Squares (GLS) regression with autocorrelation structure (corAR1), and maximum likelihood estimation using the nlme R package [60]. We measured genetic differentiation between females and males using F_{ST} measure of population differentiation [61] in vcftools version 1.16 [57].

Coalescent simulations

We modeled the expected neutral genetic diversity (π) and between-sex divergence (female vs. male F_{ST}) for the ostrich PAR following the approach of [15] and [16]. All coalescent simulations were performed using the computationally efficient simulator msprime [62]. We leveraged the fact that the coalescent for recombining sex chromosomes is mathematically equivalent to the structured coalescent for two demes (representing X and Y or Z and W chromosomes) where recombination causes migration of genes between demes. When applied to

the ostrich Z-W sex chromosomes system, the effective population sizes for the two demes corresponding to Z and W chromosomes are $3N_e/4$ and $N_e/4$, respectively, where N_e is the effective population size for an autosomal gene. The forward recombination rate in females (movement of a gene from a Z to a W chromosome) is denoted r_{f_2} while the backward rate (moving from a W to a Z chromosome) is equal to $r_{f_2}/3$ [15]. Recombination in males only shuffles genes between Z chromosomes (i.e., within deme), and therefore does not influence coalescence times. The population-scaled recombination rate was calculated as $4N_er_f$.

The key predictions from the models are the average coalescence times for genes sampled on two different Z chromosomes (\bar{T}_{ZZ}) , two W chromosomes (\bar{T}_{WW}) , and on a Z and W chromosome (\bar{T}_{ZW}) , from which we can calculate the average coalescence time for alleles sampled within a deme $(\bar{T}_w = (\bar{T}_{ZZ} + \bar{T}_{WW})/2)$ and average total coalescence time $(\bar{T}_t = \frac{(\bar{T}_{ZZ} + \bar{T}_{WW})}{4} + \frac{\bar{T}_{ZW}}{2})$. Average coalescent times for genes sampled in males and females can be calculated in similar fashion. The average genetic diversity at a given site located within the PAR will be proportional to \bar{T}_t such that, in the limit of high recombination $(r_f \rightarrow \frac{1}{2})$, \bar{T}_t for genes located in the PAR will converge on that of autosomal genes [15]. The expected genetic diversity for a site at a given location within the PAR (i.e., with N_e and r_f values determined by the physical position of a given gene) will be, approximately, $\bar{\pi}_{PAR} = \bar{\pi}_{Auto}\bar{T}_t$. The average between-sex divergence will be approximately equal to $F_{ST}^{fm} \approx 1 - \frac{\bar{T}_w^{fm}}{\bar{T}_c}$, where $\bar{T}_w^{fm} = (\bar{T}_f +$

 $\bar{T}_{mm})/2$ is the expected within-chromosome coalescence time for a pair of genes sampled in males and a pair sampled in females, and $\bar{T}_t^{fm} = (\bar{T}_{ff} + \bar{T}_{mm} + 2\bar{T}_{fm})/4$ is the expected total coalescence time for a pair of genes where one is sampled from a female and the other is sampled from a male [16].

To generate the theoretical expectations presented in Fig 4, we performed 1000 replicate coalescent simulations for each of 200 evenly located sites on the Z chromosome, starting at the PAR boundary and extending toward the distal end of the chromosome arm (encompassing all 52 MB of the PAR for analyses of the full PAR, and for only the 144 Kb immediately adjacent to the PAR boundary) using the same number of sampled chromosomes from females and males as were present in the empirical data set. For each site we used the population scaled recombination rate (ρ) as described above. Using ρ and pedigree-based recombination frequency (r), we obtained estimates of N_e across the PAR following $N_e = \frac{\rho}{h}$. We estimated recombination frequency from the genetic map obtained in [10] using the Kosambi map function $r = \frac{1}{2} \left(\frac{e^{4x} - 1}{e^{4x} + 1} \right)$, where *r* is the recombination fraction and x is the average number of crossovers. The unit of map distance measured this way is Morgan (M), one Morgan is defined as the length of a chromosome segment bracketed by two loci that produces, on average, one crossover per meiosis. For each site, we calculated $\bar{\pi}_{PAR}$ and F_{FM} from replicate simulations; confidence intervals were calculated as the 2.5 and 97.5 percentiles for each metric across replicate simulations. All computer code needed to reproduce the simulations, population genomics analyses and data processing are available in the GitHub repository (https:// github.com/Homap/ostrich_PAR_analysis).

Dryad DOI

10.5061/dryad.pnvx0k6sx.

Supporting information

S1 Table. Median read coverage per sample. (XLSX)

S2 Table. List of Z-linked scaffolds in ostrich assembly. (XLSX)

S3 Table. List of putative gametologous genes and their coordinates along the SLR Z-scaffolds and W-linked scaffolds.

(XLSX)

S4 Table. Change-point analysis for population scaled recombination rate along the Z chromosome.

(XLSX)

S1 Fig. Coverage in males (blue) and females (red) in the PAR-SLR boundary. The boundary is located on superscaffold36. Dashed lines indicate the boundary coordinates used for this study (superscaffold36: 3,516,672–3,524,264). (TIF)

S2 Fig. Distribution of alternative allele frequency (top row) and read depth per site (bottom row) for the PAR, autosomes and SLR. (TIF)

S3 Fig. Change-point analysis across the Z chromosome for population scaled recombination rate. Three significant change points were identified at 14.64, 48.1 and 53.6 Mb (red circles).

(TIF)

Acknowledgments

We thank Tim Connallon for helpful suggestions and comments on an earlier version of the manuscript.

Author Contributions

Conceptualization: Homa Papoli Yazdi, Colin Olito, Takeshi Kawakami, Charlie K. Cornwallis.

Data curation: Homa Papoli Yazdi, Per Unneberg, Mads F. Schou, Charlie K. Cornwallis.

Formal analysis: Homa Papoli Yazdi, Colin Olito, Takeshi Kawakami, Per Unneberg.

Funding acquisition: Charlie K. Cornwallis.

Investigation: Schalk W. P. Cloete, Charlie K. Cornwallis.

Methodology: Homa Papoli Yazdi, Colin Olito, Mads F. Schou, Bengt Hansson, Charlie K. Cornwallis.

Project administration: Schalk W. P. Cloete, Charlie K. Cornwallis.

Resources: Schalk W. P. Cloete.

Software: Homa Papoli Yazdi, Colin Olito.

Supervision: Charlie K. Cornwallis.

Visualization: Homa Papoli Yazdi, Colin Olito.

Writing - original draft: Homa Papoli Yazdi.

Writing – review & editing: Homa Papoli Yazdi, Colin Olito, Takeshi Kawakami, Per Unneberg, Mads F. Schou, Schalk W. P. Cloete, Bengt Hansson, Charlie K. Cornwallis.

References

- Bachtrog D, Mank JE, Peichel CL, Kirkpatrick M, Otto SP, Ashman TL, et al. Sex determination: why so many ways of doing it? PLoS Biol. 2014; 12(7):e1001899. https://doi.org/10.1371/journal.pbio.1001899 PMID: 24983465
- Kauppi L, Barchi M, Baudat F, Romanienko PJ, Keeney S, Jasin M. Distinct Properties of the XY Pseudoautosomal Region Crucial for Male Meiosis. Science. 2011; 331(6019):916–20. <u>https://doi.org/10.1126/science.1195774 PMID: 21330546</u>
- Otto SP, Pannell JR, Peichel CL, Ashman TL, Charlesworth D, Chippindale AK, et al. About PAR: The distinct evolutionary dynamics of the pseudoautosomal region. Trends Genet. 2011; 27(9):358–67. https://doi.org/10.1016/j.tig.2011.05.001 PMID: 21962971
- Bellott DW, Skaletsky H, Cho TJ, Brown L, Locke D, Chen N, et al. Avian W and mammalian Y chromosomes convergently retained dosage-sensitive regulators. Nat Genet. 2017; 49(3):387–94. <u>https://doi.org/10.1038/ng.3778 PMID: 28135246</u>
- Smeds L, Kawakami T, Burri R, Bolivar P, Husby A, Qvarnstrom A, et al. Genomic identification and characterization of the pseudoautosomal region in highly differentiated avian sex chromosomes. Nat Commun. 2014; 5:5448. https://doi.org/10.1038/ncomms6448 PMID: 25378102
- Mangs AH, Morris BJ. The human pseudoautosomal region (PAR): Origin, function and future. Curr Genomics. 2007; 8(2):129–36. https://doi.org/10.2174/138920207780368141 PMID: 18660847
- Kuhl H, Guiguen Y, Hohne C, Kreuz E, Du K, Klopp C, et al. A 180 Myr-old female-specific genome region in sturgeon reveals the oldest known vertebrate sex determining system with undifferentiated sex chromosomes. Philos Trans R Soc Lond B Biol Sci. 2021; 376(1832):20200089. <u>https://doi.org/10. 1098/rstb.2020.0089</u> PMID: <u>34247507</u>
- Han W, Liu L, Wang J, Wei H, Li Y, Zhang L, et al. Ancient homomorphy of molluscan sex chromosomes sustained by reversible sex-biased genes and sex determiner translocation. Nat Ecol Evol. 2022. https://doi.org/10.1038/s41559-022-01898-6 PMID: 36280781
- Xu L, Wa Sin SY, Grayson P, Edwards SV, Sackton TB. Evolutionary Dynamics of Sex Chromosomes of Paleognathous Birds. Genome Biol Evol. 2019; 11(8):2376–90. <u>https://doi.org/10.1093/gbe/evz154</u> PMID: <u>31329234</u>
- Yazdi HP, Ellegren H. A Genetic Map of Ostrich Z Chromosome and the Role of Inversions in Avian Sex Chromosome Evolution. Genome Biol Evol. 2018; 10(8):2049–60. https://doi.org/10.1093/gbe/evy163 PMID: 30099482
- Kosambi D. The estimation of the map distance from recombination values. Ann Eugen. 1944; 12:172– 5.
- Smith CA, Roeszler KN, Ohnesorg T, Cummins DM, Farlie PG, Doran TJ, et al. The avian Z-linked gene DMRT1 is required for male sex determination in the chicken. Nature. 2009; 461(7261):267–71. https://doi.org/10.1038/nature08298 PMID: 19710650
- Rice WR. The Accumulation of Sexually Antagonistic Genes as a Selective Agent Promoting the Evolution of Reduced Recombination between Primitive Sex-Chromosomes. Evolution. 1987; 41(4):911–4. https://doi.org/10.1111/j.1558-5646.1987.tb05864.x PMID: 28564364
- Olito C, Ponnikas S, Hansson B, Abbott JK. Consequences of partially recessive deleterious genetic variation for the evolution of inversions suppressing recombination between sex chromosomes. Evolution. 2022; 76(6):1320–30. https://doi.org/10.1111/evo.14496 PMID: 35482933
- Kirkpatrick M, Guerrero RF, Scarpino SV. Patterns of neutral genetic variation on recombining sex chromosomes. Genetics. 2010; 184(4):1141–52. https://doi.org/10.1534/genetics.109.113555 PMID: 20124026
- Kirkpatrick M, Guerrero RF. Signatures of sex-antagonistic selection on recombining sex chromosomes. Genetics. 2014; 197(2):531–41. https://doi.org/10.1534/genetics.113.156026 PMID: 24578352
- Slatkin M. Linkage disequilibrium—understanding the evolutionary past and mapping the medical future. Nat Rev Genet. 2008; 9(6):477–85. https://doi.org/10.1038/nrg2361 PMID: 18427557
- Maynard-Smith J, Haigh J. Hitch-Hiking Effect of a Favorable Gene. Genetics Research. 1974; 23 (1):23–35.
- Charlesworth B, Morgan MT, Charlesworth D. The effect of deleterious mutations on neutral molecular variation. Genetics. 1993; 134(4):1289–303. https://doi.org/10.1093/genetics/134.4.1289 PMID: 8375663

- 20. Mugal CF, Weber CC, Ellegren H. GC-biased gene conversion links the recombination landscape and demography to genomic base composition: GC-biased gene conversion drives genomic base composition across a wide range of species. Bioessays. 2015; 37(12):1317–26. <u>https://doi.org/10.1002/bies.201500058</u> PMID: 26445215
- Ratnakumar A, Mousset S, Glemin S, Berglund J, Galtier N, Duret L, et al. Detecting positive selection within genomes: the problem of biased gene conversion. Philos Trans R Soc Lond B Biol Sci. 2010; 365 (1552):2571–80. https://doi.org/10.1098/rstb.2010.0007 PMID: 20643747
- 22. Galtier N, Duret L. Adaptation or biased gene conversion? Extending the null hypothesis of molecular evolution. Trends Genet. 2007; 23(6):273–7. https://doi.org/10.1016/j.tig.2007.03.011 PMID: 17418442
- Hinch AG, Altemose N, Noor N, Donnelly P, Myers SR. Recombination in the human Pseudoautosomal region PAR1. Plos Genet. 2014; 10(7):e1004503. https://doi.org/10.1371/journal.pgen.1004503 PMID: 25033397
- Wright AE, Fumagalli M, Cooney CR, Bloch NI, Vieira FG, Buechel SD, et al. Male-biased gene expression resolves sexual conflict through the evolution of sex-specific genetic architecture. Evol Lett. 2018; 2(2):52–61. https://doi.org/10.1002/evl3.39 PMID: 30283664
- Monteiro B, Arenas M, Prata MJ, Amorim A. Evolutionary dynamics of the human pseudoautosomal regions. Plos Genet. 2021; 17(4). https://doi.org/10.1371/journal.pgen.1009532 PMID: 33872316
- Oliva M, Munoz-Aguirre M, Kim-Hellmuth S, Wucher V, Gewirtz ADH, Cotter DJ, et al. The impact of sex on gene expression across human tissues. Science. 2020; 369(6509). https://doi.org/10.1126/ science.aba3066 PMID: 32913072
- Kamiya T, Kai W, Tasumi S, Oka A, Matsunaga T, Mizuno N, et al. A Trans-Species Missense SNP in Amhr2 Is Associated with Sex Determination in the Tiger Pufferfish, Takifugu rubripes (Fugu). Plos Genet. 2012; 8(7). https://doi.org/10.1371/journal.pgen.1002798 PMID: 22807687
- Vicoso B, Emerson JJ, Zektser Y, Mahajan S, Bachtrog D. Comparative Sex Chromosome Genomics in Snakes: Differentiation, Evolutionary Strata, and Lack of Global Dosage Compensation. Plos Biology. 2013; 11(8). https://doi.org/10.1371/journal.pbio.1001643 PMID: 24015111
- Charlesworth D. Plant sex chromosome evolution. J Exp Bot. 2013; 64(2):405–20. https://doi.org/10. 1093/jxb/ers322 PMID: 23125359
- Bergero R, Qiu S, Forrest A, Borthwick H, Charlesworth D. Expansion of the Pseudo-autosomal Region and Ongoing Recombination Suppression in the Silene latifolia Sex Chromosomes. Genetics. 2013; 194(3):673–686. https://doi.org/10.1534/genetics.113.150755 PMID: 23733786
- Fridolfsson AK, Cheng H, Copeland NG, Jenkins NA, Liu HC, Raudsepp T, et al. Evolution of the avian sex chromosomes from an ancestral pair of autosomes. Proc Natl Acad Sci U S A. 1998; 95(14):8147– 52. https://doi.org/10.1073/pnas.95.14.8147 PMID: 9653155
- Yazdi HP, Ellegren H. Old but not (so) degenerated—slow evolution of largely homomorphic sex chromosomes in ratites. Mol Biol Evol. 2014; 31(6):1444–53. <u>https://doi.org/10.1093/molbev/msu101</u> PMID: 24618361
- Zhou Q, Zhang JL, Bachtrog D, An N, Huang QF, Jarvis ED, et al. Complex evolutionary trajectories of sex chromosomes across bird taxa. Science. 2014; 346(6215). <u>https://doi.org/10.1126/science.</u> 1246338 PMID: 25504727
- Tsuda Y, Nishida-Umehara C, Ishijima J, Yamada K, Matsuda Y. Comparison of the Z and W sex chromosomal architectures in elegant crested tinamou (Eudromia elegans) and ostrich (Struthio camelus) and the process of sex chromosome differentiation in palaeognathous birds. Chromosoma. 2007; 116 (2):159–73. https://doi.org/10.1007/s00412-006-0088-y PMID: 17219176
- Del Priore L, Pigozzi MI. Broad-scale recombination pattern in the primitive bird Rhea americana (Ratites, Palaeognathae). Plos One. 2017; 12(11):e0187549. https://doi.org/10.1371/journal.pone.0187549 PMID: 29095930
- **36.** Auton A, McVean G. Recombination rate estimation in the presence of hotspots. Genome Res. 2007; 17(8):1219–27. https://doi.org/10.1101/gr.6386707 PMID: 17623807
- Otto SP, Payseur BA. Crossover Interference: Shedding Light on the Evolution of Recombination. Annu Rev Genet. 2019; 53:19–44. https://doi.org/10.1146/annurev-genet-040119-093957 PMID: 31430178
- Charlesworth B. The role of background selection in shaping patterns of molecular evolution and variation: evidence from variability on the Drosophila X chromosome. Genetics. 2012; 191(1):233–46. https://doi.org/10.1534/genetics.111.138073 PMID: 22377629
- Andolfatto P, Przeworski M. Regions of lower crossing over harbor more rare variants in African populations of Drosophila melanogaster. Genetics. 2001; 158(2):657–65. https://doi.org/10.1093/genetics/ 158.2.657 PMID: 11404330

- 40. Qiu S, Bergero R, Guirao-Rico S, Campos JL, Cezard T, Gharbi K, et al. RAD mapping reveals an evolving, polymorphic and fuzzy boundary of a plant pseudoautosomal region. Mol Ecol. 2016; 25 (1):414–30. https://doi.org/10.1111/mec.13297 PMID: 26139514
- Janes DE, Ezaz T, Marshall Graves JA, Edwards SV. Recombination and nucleotide diversity in the sex chromosomal pseudoautosomal region of the emu, Dromaius novaehollandiae. J Hered. 2009; 100 (2):125–36. https://doi.org/10.1093/jhered/esn065 PMID: 18775880
- Haenel Q, Laurentino TG, Roesti M, Berner D. Meta-analysis of chromosome-scale crossover rate variation in eukaryotes and its significance to evolutionary genomics. Mol Ecol. 2018; 27(11):2477–97. https://doi.org/10.1111/mec.14699 PMID: 29676042
- Guirao-Rico S, Sanchez-Gracia A, Charlesworth D. Sequence diversity patterns suggesting balancing selection in partially sex-linked genes of the plant Silene latifolia are not generated by demographic history or gene flow. Mol Ecol. 2017; 26(5):1357–70. https://doi.org/10.1111/mec.13969 PMID: 28035715
- 44. Yazdi HP, Silva WTAF, Suh A. Why Do Some Sex Chromosomes Degenerate More Slowly Than Others? The Odd Case of Ratite Sex Chromosomes. Genes. 2020;11.
- Adolfsson S, Ellegren H. Lack of Dosage Compensation Accompanies the Arrested Stage of Sex Chromosome Evolution in Ostriches. Mol Biol Evol. 2013; 30(4):806–10. <u>https://doi.org/10.1093/molbev/</u> mst009 PMID: 23329687
- 46. Kapusta A, Suh A, Feschotte C. Dynamics of genome size evolution in birds and mammals. Proc Natl Acad Sci U S A. 2017; 114(8):E1460–E9. https://doi.org/10.1073/pnas.1616702114 PMID: 28179571
- Douglas JA, Sandefur CI. PedMine—a simulated annealing algorithm to identify maximally unrelated individuals in population isolates. Bioinformatics. 2008; 24(8):1106–8. <u>https://doi.org/10.1093/</u> bioinformatics/btn087 PMID: 18321883
- Koster J, Rahmann S. Snakemake-a scalable bioinformatics workflow engine. Bioinformatics. 2012; 28 (19):2520–2. https://doi.org/10.1093/bioinformatics/bts480 PMID: 22908215
- Martin M. Cutadapt removes adapter sequences from high-throughput sequencing reads. EMBnet. 2011; 17(1):10–2.
- Li H, Durbin R. Fast and accurate short read alignment with Burrows-Wheeler transform. Bioinformatics. 2009; 25(14):1754–60. https://doi.org/10.1093/bioinformatics/btp324 PMID: 19451168
- 51. Picard. Picard < http://broadinstitute.github.io/picard> 2018
- DePristo MA, Banks E, Poplin R, Garimella KV, Maguire JR, Hartl C, et al. A framework for variation discovery and genotyping using next-generation DNA sequencing data. Nat Genet. 2011; 43(5):491–8. https://doi.org/10.1038/ng.806 PMID: 21478889
- Quinlan AR, Hall IM. BEDTools: a flexible suite of utilities for comparing genomic features. Bioinformatics. 2010; 26(6):841–2. https://doi.org/10.1093/bioinformatics/btq033 PMID: 20110278
- Li H, Handsaker B, Wysoker A, Fennell T, Ruan J, Homer N, et al. The Sequence Alignment/Map format and SAMtools. Bioinformatics. 2009; 25(16):2078–9. <u>https://doi.org/10.1093/bioinformatics/btp352</u> PMID: 19505943
- Yazdi HP, Olito C, Kawakami T, Unneberg P, Schou M, Cloete S, et al. The evolutionary maintenance of ancient recombining sex chromosomes in the ostrich. Dryad, Dataset, <u>https://doi.org/10.5061/dryad.pnvx0k6sx</u>.
- Muggeo VM. Estimating regression models with unknown break-points. Stat Med. 2003; 22(19):3055– 71. https://doi.org/10.1002/sim.1545 PMID: 12973787
- Danecek P, Auton A, Abecasis G, Albers CA, Banks E, DePristo MA, et al. The variant call format and VCFtools. Bioinformatics. 2011; 27(15):2156–8. https://doi.org/10.1093/bioinformatics/btr330 PMID: 21653522
- Zhang C, Dong SS, Xu JY, He WM, Yang TL. PopLDdecay: a fast and effective tool for linkage disequilibrium decay analysis based on variant call format files. Bioinformatics. 2018.
- Tajima F. Statistical method for testing the neutral mutation hypothesis by DNA polymorphism. Genetics. 1989; 123(3):585–95. https://doi.org/10.1093/genetics/123.3.585 PMID: 2513255
- Pinheiro J, Bates D, DebRoy S, Sarkar D and R Core Team. nlme: Linear and Nonlinear Mixed Effects Models. 2021.
- Weir BS, Cockerham CC. Estimating F-Statistics for the Analysis of Population Structure. Evolution. 1984; 38(6):1358–70. https://doi.org/10.1111/j.1558-5646.1984.tb05657.x PMID: 28563791
- Kelleher J, Etheridge AM, McVean G. Efficient Coalescent Simulation and Genealogical Analysis for Large Sample Sizes. PLoS Comput Biol. 2016; 12(5):e1004842. <u>https://doi.org/10.1371/journal.pcbi.</u> 1004842 PMID: 27145223